Author’s response to reviews

Title: CDK1 and CCNB1 as potential diagnostic markers of rhabdomyosarcoma: Validation following bioinformatics analysis

Authors:

Qianru Li (1509094295@qq.com)
Liang Zhang (451657706@qq.com)
Jinfang Jiang (shzjjF@163.com)
Yangyang Zhang (472989983@qq.com)
Xiaomeng Wang (dream914@163.com)
Qiaochu Zhang (372193745@qq.com)
Yang Wang (499965529@qq.com)
Chunxia Liu (liuliu2239@sina.com)
Feng Li (lifeng7855@126.com)

Version: 2 Date: 29 Oct 2019

Author’s response to reviews:

Xianquan Zhan, Ph.D
Senior Editor
BMC Medical Genomics

29 Oct 2019

Dear Dr Zhan,

Re: Manuscript No. MGNM-D-19-00227R1, Submission of the revised manuscript entitled “The potential diagnostic markers of CDK1 and CCNB1 in rhabdomyosarcoma: validation following the bioinformatics analysis”

Thank you very much for your feedback. After having carefully reading your letter and checking the report, we have realized that there were multiple text overlaps in the manuscript and apologize earnestly to you for our negligence of without examining the manuscript carefully.

According to the check report, we carefully revised each overlap.
In order to prevent the same error, we used “Turnitin” repetition rate software to check the abstract, background, methods, results, discussion and conclusions of the revised manuscript. The similarity index was 3%.

Please don’t hesitate to contact us if you have any questions.

Sincerely yours,

Chunxia Liu, MD, PhD
Department of Pathology
Shihezi University School of Medicine
North 2nd Road
Shihezi, Xinjiang 832002
China
Phone: +86-138-9950-2239
E-mail: liuliu2239@sina.com